Author's response to reviews

Title: Assessment of Quality of care given to Diabetic patients at Jimma University Specialized Hospital Diabetes Follow-up clinic, Jimma, Ethiopia

Authors:

Esayas K Gudina (esakgd@gmail.com)
Solomon Tamiru (solomontamiru@yahoo.com)
Fessahaye Alemseged (fessahayeatd@yahoo.com)
Rana Ram (usctrojan.rana@gmail.com)

Version: 7 Date: 16 December 2011

Author's response to reviews:

Authors details on the title page has been revised based on the comment by editorials.
Corresponding author is also indicated.
Table shading and alignments have been changed according to the recommendation.
Few typographical errors have been corrected.
We all grateful for all your efforts and that of the peer reviewers. If you need to modify the manuscript for editorial purpose, it is welcome.